Author's response to reviews

Title: Changing clinical needs of people living with AIDS and receiving home based care in Malawi - the Bangwe Home Based Care Project 2003-2008 - a descriptive study

Authors:

Cameron Bowie (cam.bowie1@gmail.com)
Norton Gondwe (nortongondwe@yahoo.com)
Claire Bowie (claire.bowie1@gmail.com)

Version: 5 Date: 21 June 2010

Author's response to reviews:

1. Author details in manuscript changed to be the same as author details here
2. Tables reformatted as required
3. Discussion changed to conclusions in abstract
4. Copyright line removed
5. Figure 4 recropped to correct size and in same way as other figures but looks different when uploaded. Please deal with this.

All requested changed done and manuscript checked. Thanks.